Author's response to reviews

Title: 'Sepsis-related anemia' is absent at hospital presentation; a retrospective cohort analysis.

Authors:

Geertje Jansma (jansma1987@gmail.com)
Fellery de Lange (Fellery.de.Lange@ZNB.NL)
W Peter Kingma (W.P.Kingma@ZNB.NL)
Namkje AR Vellinga (namkjevellinga@hotmail.com)
Matty Koopmans (Matty.Koopmans@ZNB.NL)
Michael A Kuiper (M.Kuiper@ZNB.NL)
E Christiaan Boerma (e.boerma@chello.nl)

Version: 6 Date: 15 December 2014

Author's response to reviews:

Dear editor,

Thank you for the opportunity to improve the quality of our paper entitled: ‘Sepsis-related anemia’ is absent at hospital presentation; a retrospective cohort analysis. We thank the reviewers for the valuable comments. We have answered the comments one by one in a separate document and depicted the changes in the text of the revised version in red. Text editing was performed by a professional. An independent statistician (N. Veeger) confirmed the robustness of the statistical methods that we applied. As a result of this review process the quality of the paper considerably improved.

We hope for a positive result.

Kind regards,

Geertje Jansma